% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P4.15}{23(4)}{2016}
% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need
% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}
\newtheorem{result}[theorem]{Result}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}
\newcommand{\jvec}{\mbox{\boldmath $\jmath$}}
%CAL-commands
\newcommand{\CA}{\mbox{$\cal A$}}\newcommand{\CB}{\mbox{$\cal B$}}
 \newcommand{\CC}{\mbox{$\cal C$}}\newcommand{\CD}{\mbox{$\cal D$}}
 \newcommand{\CE}{\mbox{$\cal E$}}\newcommand{\CF}{\mbox{$\cal F$}}
 \newcommand{\CG}{\mbox{$\cal G$}}\newcommand{\CH}{\mbox{$\cal H$}}
 \newcommand{\CI}{\mbox{$\cal I$}}\newcommand{\CJ}{\mbox{$\cal J$}}
 \newcommand{\CK}{\mbox{$\cal K$}}\newcommand{\CL}{\mbox{$\cal L$}}
 \newcommand{\CM}{\mbox{$\cal M$}}\newcommand{\CN}{\mbox{$\cal N$}}
 \newcommand{\CO}{\mbox{$\cal O$}}\newcommand{\CP}{\mbox{$\cal P$}}
 \newcommand{\CQ}{\mbox{$\cal Q$}}\newcommand{\CR}{\mbox{$\cal R$}}
 \newcommand{\CS}{\mbox{$\cal S$}}\newcommand{\CT}{\mbox{$\cal T$}}
 \newcommand{\CU}{\mbox{$\cal U$}}\newcommand{\CV}{\mbox{$\cal V$}}
 \newcommand{\CW}{\mbox{$\cal W$}}\newcommand{\CX}{\mbox{$\cal X$}}
 \newcommand{\CY}{\mbox{$\cal Y$}}\newcommand{\CZ}{\mbox{$\cal Z$}}
 \newcommand{\G}{\overline{G}}
 \newcommand{\R}{\mathbb{R}}
\newcommand{\AT}{\mathbb{ATLAS}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\Q}{\mathbb{Q}}
%\newcommand{\C}{\mathbb{C}}
\newcommand{\C}{\mathcal{C}}
\newcommand{\F}{\mathbb{F}}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf Some self-orthogonal codes\\ related to Higman's geometry}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{Jamshid Moori\thanks{Support of the National Research Foundation
of South Africa and North-West University are acknowledged.}\\
\small School of Mathematical Sciences \\[-0.8ex]
\small North-West University (Mafikeng) \\[-0.8ex]
\small  Mmabatho 2735, South Africa\\
\small\tt Jamshid.Moori@nwu.ac.za\\
\and
B.~G. Rodrigues\thanks{This work is based on the research supported
by the National Research Foundation
of South Africa (Grant Numbers 87470 and 91495).} \\
\small School of Mathematics, Statistics and Computer Science\\[-0.8ex]
\small University of KwaZulu-Natal \\[-0.8ex]
\small Durban 4000, South Africa\\
\small\tt rodrigues@ukzn.ac.za}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Sep 4, 2015}{Oct 13, 2016 }{Oct 28, 2016}\\
\small Mathematics Subject Classifications: 05B05, 20D08, 94B05}

\begin{document}
\begin{thebibliography}{99}

\bibitem{ak:book}
E.~F. Assmus, Jr and J.~D. Key.
\newblock {\em Designs and their Codes}.
\newblock Cambridge: Cambridge University Press, 1992.
\newblock Cambridge Tracts in Mathematics, Vol.~103 (Second printing with
  corrections, 1993).

\bibitem{magma}
W. Bosma, J. Cannon, C. Playoust.
\newblock The Magma algebra system I: The user language.
\newblock {\em J. Symbolic Comput.} \textbf{24} (1997), 235--265.

\bibitem{caldbr}
A.~R. Calderbank and D.~B. Wales.
\newblock A global code invariant
under the {H}igman-{S}ims group.
\newblock  \emph{J. Algebra}, \textbf{75} (1982), 233--260.

\bibitem{atlas}
J.H. Conway, R.T. Curtis, S.P. Norton, R.A. Parker, and R.A. Wilson.
\newblock \emph{{A}tlas of {F}inite {G}roups}.
\newblock Oxford University Press, Oxford, 1985.

\bibitem{connor}
T.~ Connor and D.~Leemans.
\newblock An atlas of subgroup lattices of finite almost simple groups.
\newblock {\em Ars Math. Contemp.}, \textbf{8}, (2015), 259--266.


\bibitem{hs2sublattice}
T.~ Connor and D.~Leemans.
\newblock An atlas of subgroup lattices of finite almost simple groups, 2014.
\newblock http://homepages.ulb.ac.be/~tconnor/atlaslat/hsd2.pdf accessed July 2015.

\bibitem{crmikua}
D.~Crnkovi{\'c} and V. Mikuli{\'c}.
\newblock Unitals, projective planes and other combinatorial structures constructed from the unitary groups ${U}_3(q)$
$q=3,4,5,7.$
\newblock {\em Ars Combin.}, \textbf{110}, (2013), 3--13.

\bibitem{crovmikrod}
\newblock ~D. Crnkovi{\'c} and V. Mikuli{\'c}, and B. G. Rodrigues.
\newblock Designs, strongly regular graphs and codes constructed from some primitive groups.
\newblock  In {\em Information Security, Coding Theory and Related Combinatorics} (D. Crnkovi{\'c} \& V. Tonchev, Eds.), IOS Press, Amsterdam,
 2011, pages 231-252.

\bibitem{Higgeom}
G.~Higman.
\newblock On the simple group of D. G. Higman and C.~C.~ Sims.
\newblock {\em Illinois J. Math.}, \textbf{13} (1969), 74--80.

\bibitem{km}
J.~D.~Key and J.~Moori, Codes.
\newblock  Designs and Graphs from the Janko Groups $J_1$ and $J_2$,
\newblock {\em  J. Combin. Math. Combin. Comput.}, \textbf{40} (2002), 143--159.

\bibitem{km1}
J.~D.~Key and J.~Moori.
\newblock Correction to: Codes, designs and graphs from the Janko groups $J_1$ and $J_2$ [J. Combin. Math. Combin. Comput. 40
(2002), 143-159].
\newblock {\em J. Combin. Math. Combin. Comput.}, \textbf{64} (2008), 153.

\bibitem{key1}
J.~D.~Key and J.~Moori, B. G. Rodrigues.
\newblock On some designs and codes from primitive representations of some finite simple groups.
\newblock {\em J. Combin. Math. Combin. Comput.}, \textbf{ 45} (2003), 3--19.

\bibitem{knapp&schaeffer}
W.~ Knapp and H.~-J. Schaeffer.
\newblock On the codes related to the Higman-Sims graph.
\newblock {\em Electron. J. Combin.}, \textbf{22} (1) (2015), \#P1.19.

\bibitem{magl}
S.~S. Magliveras.
\newblock The {S}ubgroup {S}tructure of the {H}igman-{S}ims   {S}imple group.
\newblock {\em Bull. Amer. Math. Soc.}, \textbf{77} (1971), no.~4, 535--539.

\bibitem{moorod8}
J.~Moori and B.~G. Rodrigues.
\newblock On some designs and codes invariant under the {H}igman-{S}ims group.
\newblock {\em Util. Math.}, \textbf{86} (2011), 225--239.


\bibitem{higmanternary}
B.~G. Rodrigues.
\newblock Some new $2$-designs and complementary dual codes related to the Higman-Sims group.
\newblock Submitted.

\bibitem{colva-aff1000}
C. M. Roney-Dougal  and W. R. Unger.
\newblock The affine primitive permutation groups of degree less than 1000.
\newblock  {\em J. Symbolic Comput.}, \textbf{35} (2003), no. 4, 421--439.

\bibitem{tonch4}
V.~D. Tonchev.
\newblock Binary codes derived from the {H}offman-{S}ingleton and {H}igman-{S}ims graphs.
\newblock {\em IEEE Trans. Info. Theory}, \textbf{43} (1997), 1021--1025.

\end{thebibliography}


\maketitle

% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
 We examine some self-orthogonal codes constructed from a rank-5 primitive permutation representation of degree 1100 of the sporadic
simple group ${\rm HS}$ of Higman-Sims. We show that Aut($C) = {\rm HS}{:}2$, where $C$ is a code of dimension 21 associated with Higman's geometry.

% keywords are optional
  \bigskip\noindent \textbf{Keywords:} Linear codes; Higman's geometry; Higman-Sims group
\end{abstract}

\section{Introduction}\label{intro}

The study of binary linear codes invariant under the Higman-Sims
group (${\rm HS}$), in particular those constructed from the
primitive permutation representations of degrees 100, 176 and 1100
respectively has been carried in \cite{caldbr,tonch4} and
in \cite{moorod8}. Recently, Knapp and Schaeffer \cite{knapp&schaeffer} using representation theoretic methods provided an elegant account on the binary codes of length 100 related with the Higman-Sims graph. In the paper \cite{higmanternary}, the second author offers an account on non-binary codes from the representations of degree 100 and constructs new $2$-designs from the representation of degree 176. It is well
known that the Higman-Sims group possesses two  inequivalent rank-5
primitive representations of degree $1100$, one on the set of edges
of $\CG$, the Higman-Sims graph with parameters $(100,22,0,6)$, with
edge stabilizer isomorphic to $L_3(4){:}2_1$, and the other on the
set of conics of G. Higman's geometry (see \cite{atlas,Higgeom})
with stabilizer of a conic isomorphic to $S_8$. The orbits of the
action on the cosets of $L_3(4){:}2_1$ have lengths $ 1, 42, 105,
280 $ and $672$ respectively, while those of the action on the
cosets of $S_8$ have lengths $1, 28, 105, 336$ and $630.$ In
\cite{moorod8}, using the orbit of length 672  we constructed the
unique and minimal degree faithful irreducible $\F_2$-representation
(20-dimensional) of the Higman-Sims group, as a binary $[1100, 20,
480]_2$ code. A review of our paper [On some designs and codes
invariant under the Higman-Sims group' (Util. Math., 2011),
MR2884789 (2012m:05082)] prompted us to examine the extent of a possible relation between the 20-dimensional code constructed in \cite{moorod8} and Higman's geometry. The careful reader will notice that Higman's
geometry originates from an action on the cosets of $S_8$ and not on
the cosets of $L_3(4){:}2_1$. Due to this, it would seem that no relation could be borne
between the said 20-dimensional code and Higman's geometry. However, on examining the question on the existence of binary codes related with the geometry of G. Higman we were able to show that the  20-dimensional code referred to above is a subcode of codimension 1 in a $[1100,21, 420]_2$ code constructed in this paper and related to Higman's geometry.
To deal with this question we use a method proposed in \cite{crmikua}, and construct a self-dual
symmetric $1$-$(1100,420, 420)$ design $\CD$ taking for point set
the conjugacy classes of a maximal subgroup isomorphic to
$L_3(4){:}2_1$ and for block set the conjugacy classes of a maximal
subgroup isomorphic to $S_8$. The binary row span of the incidence
matrix of $\CD$ induces a 21-dimensional $[1100, 21, 420]_2$ code
whose properties we examine in the sequel.\\
The paper is organized as follows: in Section~\ref{sec-tan} we
outline our notation and give a brief overview of the ${\rm HS}$
group in Section~\ref{sec-hs}. In Section~\ref{construction} we
describe the construction method used and in Section~\ref{sec-codes}
and Section~\ref{compdescode} we present our results.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Terminology and
notation}\label{sec-tan}

Our notation will be standard, and it is as in \cite{ak:book} and
$\AT$ \cite{atlas}. For the structure of groups and their maximal
subgroups we follow the $\AT$ notation. The groups $G.H,$ $G{:}H,$
and $G^{\cdot}H$ denote a general extension, a split extension and a
non-split extension respectively. For a prime $p,$ the symbol $p^m$
denotes an elementary abelian group of that order. The notation
$p_{+}^{1+2n}$ and $p_{-}^{1+2n}$ are  used for extraspecial groups
of order $p^{1+2n}$. If $p$ is an odd prime, the subscript is $+$ or
$-$ according as the group has  exponent $p$ or $p^2.$ For $p=2$ it
is $+$ or $-$ according as the central product has an even or odd
number of quaternionic factors.

An incidence structure $\CD =( \CP,\CB,\CI)$, with point set \CP,
block set \CB\ and incidence \CI\ is a $t$-$(v,k,\lambda)$ design,
if $|\CP|=v$, every block $B \in \CB$ is incident with precisely $k$
points, and every $t$ distinct points are together incident with
precisely $\lambda$ blocks. The {\bf complement} of $\CD$ is the
structure $\tilde{\CD} =( \CP,\CB,\tilde{\CI}),$ where
$\tilde{\CI}=\CP\times\CB - \CI.$ The {\bf dual} structure of $\CD$
is $\CD^t =(\CB,\CP,\CI^t),$ where $(B,p)\in \CI^t$ if and only if
$(P,B)\in \CI$. Thus the transpose of an incidence matrix for \CD\
is an incidence matrix for $\CD^t$. We will say that the design is
{\bf symmetric} if it has the same number of points and blocks, and
{\bf self dual} if it is isomorphic to its dual.

The code $C_F$ of the design \CD\ over the finite field $F$ is the
space spanned by the incidence vectors of the blocks over $F$. We
take $F$ to be a prime field $F_p$, in which case we write also
$C_p$ for $C_F$, and refer to the dimension of $C_p$ as the {\bf
$p$-rank} of \CD.

If the point set of \CD\ is denoted by \CP\ and the block set by
\CB, and if \CQ\ is any subset of \CP, then we will denote the
incidence vector of \CQ\ by $v^{\cal Q}$.
 Thus $C_F = \left< v^B \,| \, B \in \CB \right>$,
and is a subspace of $F^{\cal P}$, the full vector space of
functions from \CP\ to $F$. The dual or {\bf orthogonal} code
${C_F}^\perp$ of $C_F$ is the orthogonal subspace under the standard
inner product. The {\bf hull} of a design's code over some field is
the intersection $C_F \cap {C_F}^\perp$.
 If a linear code over a field of order $q$
is of length $n$, dimension $k$, and minimum weight $d$, then we
write $[n,k,d]_q$ to represent this information.  A {\bf constant
word} in the code is a codeword all of whose coordinate entries are
the same. The all-one vector will be denoted by \jvec, and is the
constant vector of weight the length of the code. Two linear codes
of the same length and over the same field are {\bf equivalent} if
each can be obtained from the other by permuting the coordinate
positions and multiplying each coordinate position by a non-zero
field element. They are {\bf isomorphic} if they can be obtained
from one another by permuting the coordinate positions. An {\bf
automorphism} of a code is any permutation of the coordinate
positions that maps codewords to codewords. An automorphism thus
preserves each weight class of $C.$


\section{The Higman-Sims group and its automorphism group}\label{sec-hs}

As we had in \cite{moorod8}, the Higman-Sims simple group can be
constructed from the Higman-Sims graph $\CG$. Let $\CG = (\Omega ,
\CE)$ be a graph of valence $22$ on the set $\Omega$ of $100$ points
such that any given vertex has $22$ neighbours (points) and the
remaining $77$ vertices are joined to $6$ of these points and may be
labelled by the corresponding hexad. Two of the $77$ vertices are
joined only if the corresponding hexads are disjoint.  The hexads
form a Steiner system $S(3,6,22)$. The Higman-Sims simple group
${\rm HS}$ is the subgroup of even permutations of Aut$(\CG) \cong
{\rm HS}{:}2$, the automorphism group of ${\rm HS}$. The point
stabilizer of Aut$(\CG)$ on $\Omega$ is Aut$(S(3,6,22)) \cong
M_{22}{:}2$ and the order of the Higman-Sims group ${\rm HS}$ is
$44352000 = 2^9{\cdot}3^2 {\cdot}5^3{\cdot}7{\cdot}11$.
 The group ${\rm HS}$ has two inequivalent representations of degree $1100$, one on the
set of edges of $\CG$ with point stabilizer isomorphic to
$L_3(4){:}2_1$ and the other on the set of conics of G. Higman's
geometry (see \cite{atlas}) with point stabilizer isomorphic to
$S_8$. The subgroup $S_8$ is also the set stabilizer of a fixed
outer automorphism of ${\rm HS}$.

\newpage

\begin{result}(Magliveras \cite{magl})\label{mag} $\mbox{}$
The Higman-Sims group ${\rm HS}$ has exactly $12$ conjugacy classes
of maximal subgroups, as follows:
\[ \begin{array}{ll}
 M_{22} & U_3(5){:}2 \quad (2 \ classes) \\
L_3(4){:}2_1 & S_8 \\
2^4.S_6 & 4^3{:}L_3(2) \\
M_{11} \quad (2 \ classes) & 4{\cdot}2^4{:}S_5  \\
2 \times A_6{\cdot}2^2 & 5{:}4 \times A_5.
\end{array} \]
\end{result}

The primitive representations described in Result \ref{mag} are of
degrees 100, 176, 176, 1100, 1100, 3850, 4125, 5600, 5600, 5575,
15400 and 36960 respectively. In TABLE~1 below the first column
depicts the ordering of the primitive representations of ${\rm HS}$ and ${\rm HS}{:}2$ respectively, as given by
Magma (or the $\AT$~\cite{atlas}) and as used in our computations;
the second gives the maximal subgroups; the third gives the degrees
(the number of cosets of the point stabilizer).
%TABLE~2 gives the
%same information for ${\rm HS}{:}2$ that TABLE~1 provides for ${\rm
%HS}$.

\hspace{.5cm}
%\flushleft

%\hbox to \hsize{%
%\begin{minipage}[b]{0.5\linewidth}
%\centering {\footnotesize
\begin{center}
{\footnotesize
TABLE~1: Maximal subgroups of ${\rm HS}$ and ${\rm HS}{:}2$\\[1ex]
\begin{tabular}{|c|c|c||c|c|} \hline
No. & Max. sub. of ${\rm HS}$ & Deg. & Max. sub. of ${\rm HS}{:}2$ & Deg.\\ \hline
\hline 1&$M_{22}$ & 100 &${\rm HS}$ & 2\\
\hline 2&$U_3(5):2$ & 176 &$M_{22}:2$ & 100 \\
\hline 3&$U_3(5):2$& 176 &$L_3(4){:}2^2$& 1100 \\
\hline 4&$L_3(4):2_1$ & 1100 &$S_8 \times 2$ &1100\\
\hline 5&$S_8$ & 1100 &$2^5 \cdot S_6$ & 3850\\
\hline 6&$2^4 \cdot S_6$ & 3850& $4^3{:}(L_3(2) \times 2)$ & 4125 \\
\hline 7&$4^3:L_3(2)$ & 4125 &$2_{+}^{1 +6} {:} S_5$ & 5775 \\
\hline 8&$M_{11}$ & 5600&$2 \times A_6 \cdot 2^2\cdot 2$ & 15400\\
\hline 9&$M_{11}$& 5600 &$5^{1+2} {:}(Q_8{:}4)$ & 22176 \\
\hline 10&$4^{.}2^4 {:} S_5$ & 5775 & $ 5{:} 4 \times S_5$ & 36960\\
\hline 11&$2\times A_6.2^2$ & 15400 &&\\
\hline 12&$5:4\times A_5$ &36960 &&\\
\hline \hline
\end{tabular}}
\end{center}
%\end{minipage}}
%\hfill
%\hspace{0.2cm}
%\begin{minipage}[b]{0.5\linewidth}
%\centering {\footnotesize
%TABLE~2: Max subgroups of ${\rm HS}{:}2$\\[1ex]
%\begin{tabular}{|c|c|c|} \hline
%No. & Max. sub. & Deg. \\ \hline
%\hline 1&${\rm HS}$ & 2 \\
%\hline 2&$M_{22}:2$ & 100 \\
%%\hline 3&$5_{+}^{1+2} {:} [2^5]$& 176  \\
%\hline 3&$L_3(4){:}2^2$& 1100  \\
%\hline 4&$S_8 \times 2$ &1100 \\
%\hline 5&$2^5 \cdot S_6$ & 3850\\
%\hline 6& $4^3{:}(L_3(2) \times 2)$ & 4125  \\
%\hline 7&$2_{+}^{1 +6} {:} S_5$ & 5775 \\
%\hline 8&$2 \times A_6 \cdot 2^2\cdot 2$ & 15400\\
%\hline 9&$5^{1+2} {:}(Q_8{:}4)$ & 22176\\
%\hline 10& $ 5{:} 4 \times S_5$ & 36960 \\
%\hline \hline
%\end{tabular}}
%\end{minipage}}


\section{The construction}\label{construction}
Crnkovi{\'c} and Mikuli{\'c} in \cite{crmikua} (see also
\cite{crovmikrod}) gave a method that outlines a construction of
$1$-designs from finite permutation groups, which are not
necessarily symmetric, and stabilizers of a point and a block that
are not necessarily conjugate. This result is a generalization of an
earlier construction of symmetric 1-designs and regular graphs which
was described in \cite[Proposition~1]{km}, used in \cite{key1} and
later corrected in~\cite{km1}. For the sake of completeness and
readiness of use we state the result below.

\begin{result}
\label{tm:main} Let $G$ be a finite permutation group acting
primitively on the sets $\Omega_1$ and $\Omega_2$ of size $m$ and
$n$, respectively. Let $\alpha \in \Omega_1$ and $\delta \in
\Omega_2$ and let $\Delta_2 = \delta G_{\alpha}$ be the
$G_{\alpha}$-orbit of $\delta \in \Omega_2$ and $\Delta_1 = \alpha
G_{\delta}$ be the $G_{\delta}$-orbit of $\alpha \in \Omega_1.$  If
$\Delta_2 \neq \Omega_2$ and ${\mathcal{B}}=\{ \Delta_2 g : g \in G
\},$ then $\CD(G, \alpha, \delta) = (\Omega_2,{\mathcal{B}})$ is a
$1$-$(n, | \Delta_2 |, |\Delta_1|)$ design with $m$ blocks, and $G$
acts as an automorphism group, primitive on points and blocks of the
design.
\end{result}

\begin{remark}
\label{rmk1}
Let  $M_1$ and $M_2$ to be two maximal subgroups of a finite group $G$. We denote the
conjugacy class of $M_i,\ i=1,2,$ by $ccl_{G}(M_i)$ and
$|ccl_{G}(M_i)|=[G:N_G(M_i)]$. The elements of $ccl_{G}(M_i),\ i=1,2,$ are denoted by $M_i^{g_1}, M_i^{g_2}, \ldots, M_i^{g_{j_i}}$ and thus we obtain $j_i=[G:N_G(M_i)]$.\\
Observe that $G$ acts primitively on $ccl_G(M_1)$ and $ccl_G(M_2)$
by conjugation. In this way a primitive $1-$design can be
constructed such that:
\begin{itemize}
 \item the point set of the design is $ccl_{G}(M_2)$, and the block set is $ccl_{G}(M_1)$,
 \item the block $M_1^{g_i}$ is incident with the point $M_2^{h_j}$ if and only if
 $M_1^{g_i} \cap M_2^{h_j} \cong M_1 \cap M_2.$
\end{itemize}
In the case when $G$ is a finite simple group and $M_1$ and $M_2$ are two maximal subgroups of $G$, then clearly  $N_G(M_i)=M_i$  and hence $j_i=[G:M_i]$ for $i= 1, 2.$
\end{remark}

\section{A $[1100, 21, 420]_2$ code invariant under ${\rm HS}{:}2.$}\label{sec-codes}
Recall that our interest is in the construction of designs and codes
which bear an association with Higman's geometry. To this end,
we consider the description given in Section~\ref{sec-hs} for the
Higman-Sims group and the discussion by G.~Higman in
\cite[Section~1, p.75]{Higgeom} (see also \cite[Sections~4, and 5]{knapp&schaeffer} for a model of Higman's geometry) to construct Higman's geometry.
Hence, taking for points the edges of the graph $\CG$ and for blocks
the conics of the geometry we construct a $1$-$(1100, 420, 420)$ design $\CD$  on which
${\rm HS}$ acts primitively on points and on blocks. It will be intuitive to notice that if we use Result~\ref{tm:main} and
Remark~\ref{rmk1} we take for point set $\Omega_2$ the conjugacy
classes of a maximal subgroup isomorphic to $L_3(4){:}2_1$ and for
block set $\Omega_1$ the conjugacy classes of a maximal subgroup
isomorphic to $S_8$. Notice that $\Omega_1$, and $\Omega_2$ are
primitive ${\rm HS}$-sets of degree 1100. In the sequel we examine
the properties of a binary $[1100, 21, 420]_2$ self-orthogonal and
doubly-even code $C$ determined by the row span of the incidence
matrix of $\CD$ and explore its possible connections with
Higman's geometry.

\begin{lemma} \label{thedesign}
Let $G = {\rm  HS}{:}2$ and let $\CD = (\Omega_2, \CB)$ be a design constructed as in Result~\ref{tm:main} taking for point set $\Omega_2$ the conjugacy
classes of a maximal subgroup isomorphic to $L_3(4){:}2_1$ and for block set $\Omega_1$ the conjugacy classes of a maximal subgroup isomorphic to $S_8$. Then $\CD$ is a self-dual, symmetric $1$-$(1100,420,420)$ design with $G = {\rm Aut}(\CD)$ acting point- and block-primitively.
\end{lemma}

\begin{proof}From Result~\ref{tm:main} it is clear that $G \subseteq {\rm
Aut}(\CD)$. Once again, from Result~\ref{tm:main}, and also from the
$\AT$~\cite[p.80]{atlas} we see that $G$ acts primitively on both
$\Omega_2$ and $\Omega_1$, where $\Omega_1$ and $\Omega_2$ represent
the point and block sets of $\CD$ and these are the edges of the
Higman-Sims graph $\CG$ and the conics of Higman's geometry,
respectively, (in terms of Result~\ref{tm:main} these are the sets
of conjugacy classes of a maximal subgroup isomorphic to
$L_3(4){:}2_1$ and of a maximal subgroup isomorphic to $S_8$
respectively) with degree $|\Omega_1| = |\Omega_2| = 1100.$ This shows that $\CD$ is a point primitive, symmetric $1$-design. Moreover, the
stabilizers $G_x$ and $G_{B}$ of a point $x \in \Omega_2$ and of a
block $B \in \CB$ have five orbits, namely $\Phi_1 = \{x\}$, $\Phi_2$, $\Phi_3$,
$\Phi_4$, and $\Phi_5$ with subdegrees:1, 42, 105, 280, and 672; and three orbits
 namely $\Psi_1$, $\Psi_2$ and $\Psi_3$ with subdegrees: 120, 420 and 560.
 It remains to show that $G =
{\rm Aut}(\CD).$ Now $G \subseteq {\rm Aut}(\CD)\subseteq S_{1100},$ so ${\rm Aut}(\CD)$ is a primitive permutation group on $\Omega_2$
of degree 1100. Moreover, ${\rm Aut}(\CD)_{x}$ must fix $\Delta_2$ setwise, and hence ${\rm Aut}(\CD)_{x}$ also has orbits of lengths
1, 42, 105, 280, and 672 in $\Omega_2$. The only primitive group of degree 1100, such that ${\rm Aut}(\CD)_{x}$ has orbit lengths 1, 42, 105, 280, and 672 is ${\rm HS{:}2}$, see
\cite[Table~9,p.178]{colva-aff1000}. Hence $G = {\rm Aut}(\CD).$
~\end{proof}

Taking the binary row span of the incidence matrix of $\CD$ we
obtain the 21-dimensional ${\rm HS}{:}2$-invariant $[1100,21,420]_2$
code whose properties are discussed in Proposition~\ref{thecodecode} below.

\begin{proposition}\label{thecodecode}
Let $C$ be the binary code defined by the incidence matrix of $\CD$.
Then $C$ is a self-orthogonal doubly-even $[1100,21,420]_2$ code.
Its dual code ${C}^\perp$ is a $[1100,1079,4]_2$ with words of
weight $4$. Furthermore, $\jvec \in {C}^\perp$ and {\rm Aut}$(C)
\cong {\rm HS}{:}2.$
\end{proposition}

\begin{proof}The parameters of $C$ were determined through computations with
Magma~\cite{magma}. Since the dimension of the $C$ equals the
dimension of $C \cap {C}^{\perp}$, we have $C
\subseteq C^{\perp}$ and so $C$ is self orthogonal.
Since ${\rm HS}$ is a normal subgroup of ${\rm HS}{:}2$ it follows that $C$ is
${\rm HS}{:}2$-invariant.

Notice from TABLE~3 that there are exactly 1100 codewords of minimum weight 420
in $C$. Thus the minimum weight codewords are the incidence
vectors of the blocks of the design $\CD$, and hence spanning
vectors of $C$. From this we deduce that ${\rm Aut}(\CD) \subseteq {\rm Aut}(C)$. Now, order considerations shows that ${\rm Aut}(C) \cong {\rm HS}{:}2.$
Furthermore, since the spanning words of $C$ have weight
divisible by four, it follows that $C$ is doubly-even.

\medskip
The weight distribution of $C$ is listed in TABLE~3. In TABLE~3, $l$
represents the weight of a codeword and $A_l$ denotes the number of
codewords of weight $l.$

\begin{center}
{\footnotesize
TABLE~3: The weight distribution of $C$\\[1ex]
\begin{tabular}{lr||lr}
\hline \hline $l$ & $\qquad\qquad\quad\;\;\, A_l$ & $l$ & $\qquad\qquad\quad\;\;\, A_l$\\
\hline
0    & 1    & 544 &  793100 \\[1ex]
420  & 1100  & 548 &  500500\\[1ex]
480  & 15400 & 564 & 308000 \\[1ex]
484  & 100   & 576 &  231000  \\[1ex]
500  & 22176  & 612 &  23100 \\[1ex]
512 & 7975 & 672 &   1100 \\[1ex]
532 & 193600 &   &  \\[1ex]
\hline \hline
\end{tabular}}
\end{center}

In addition note that the blocks of $\CD$ are of even size, so $\jvec$ meets evenly every vector of $C,$  and thus $\jvec \in {C}^{\perp}$.
Finally, using MacWilliams identities and Pless' power moment identities we obtain the minimum weight 4 for ${C}^{\perp}.$ ~\end{proof}

\begin{remark}
The code and designs found above can be described geometrically: the
1100 codewords of weight 420 are the incidence vectors of the blocks
of the design $\CD$,  and these represent the conics of Higman's
geometry.
\end{remark}

\subsection{Stabilizer in ${\rm HS}{:}2$ of a word of weight
$l$}\label{stabhs2}

Let $L =\{420, 480, 484, 500, 512,\,672\}$ and $\overline{L}
=\{532,\, 544, 548, 564, 576, 612\}$. For $l \in  L \cup
\overline{L}$ we define $W_l = \{ w_l \in C \;|\; {\rm wt}(w_l) =
l\}.$ Since ${\rm Aut}(C) \cong {\rm HS}{:}2$, in this section we
determine the structures of the stabilizers $({\rm HS}{:}2)_{w_l}$,
for all nonzero weight $l$.

We show in Lemma~\ref{maxims} that for $l \in L$ the stabilizer
$({\rm HS}{:}2)_{w_l}$ is a maximal subgroup of ${\rm HS}{:}2,$
where $({\rm HS}{:}2)_{w_{420}}\cong S_8 \times 2,\,\,({\rm
HS}{:}2)_{w_{480}}\cong 2 \times A_6 \cdot 2^2\cdot 2,\,\,\,\,({\rm
HS}{:}2)_{w_{484}}\cong M_{22}{:}2,\,\, ({\rm
HS}{:}2)_{w_{500}}\cong 5^{1+2} {:}(Q_8{:}4),\,\, ({\rm
HS}{:}2)_{w_{512_1}}\cong 2^5 \cdot S_6,\, ({\rm
HS}{:}2)_{w_{512_2}}\cong 4^3{:}(L_3(2) \times 2)
\text{~and~}\,\\({\rm HS}{:}2)_{w_{672}}\cong L_3(4){:}2^2.$ Now for
$w_l \in W_l$ we take the support of $w_l$ and orbit it under ${\rm
HS}{:}2$ to form the blocks of the $1$-$(1100,l,k_l)$ designs
$\CD_{w_l}$, where $k_l=|(w_l)^{{\rm HS}{:}2}|\times
\dfrac{l}{1100}$. We show that for all $l\in L$, ${\rm HS}{:}2$ acts
primitively on these designs. Information on these designs is given
in TABLE 4 and TABLE 5.

Next in Lemma~\ref{notmaxims} by considering $w_l$ where $l \in
\overline{L}$ we describe the structures of $({\rm HS}{:}2)_{w_l}$
and show that these are not maximal in ${\rm HS}{:}2.$
\begin{lemma}\label{maxims}
Let  $l \in L$ and $w_l \in W_l.$ Then $({\rm HS}{:}2)_{w_l} $ is a
maximal subgroup of ${\rm HS}{:}2.$ Furthermore ${\rm HS}{:}2$ is
primitive on ${\CD}_{w_l}\ $ for each $l.$
\end{lemma}
\begin{proof}First assume that $l \in \{420, 480, 484, 500,\, 672\}.$  Since ${\rm HS}$ is transitive on $W_l$, so is ${\rm HS}{:}2$.
Hence for $l \in L,$ each $W_l$ forms an orbit under the action of
${\rm HS}{:}2$, so that $({\rm HS})_{w_l}$ is subgroup of index 2 in
$({\rm HS}{:}2)_{w_l}.$ Therefore by the orbit stabilizer Theorem
and the $\AT$ (or right hand side of TABLE~1) we have $[{\rm HS} {:}2\,{:}\,({\rm
HS}{:}2)_{w_l}] \in \{1100, 15400, 100, 22176, 4125, 3850\}.$
Using the list of maximal subgroups of $HS{:}2$ (see right hand side of TABLE~1), we
deduce that $({\rm HS}{:}2)_{w_{420}} \in \{L_3(4){:}2^2,\; S_8
\times 2, \:S\}$, where $S$ possibly is a subgroup of $M_{22}{:}2$
of index $11$. Examining the list of maximal subgroups of
$M_{22}{:}2$ in \cite{atlas} or \cite{connor}, we can easily see
that $M_{22}{:}2$ contains no subgroup of index $11$. Also direct
calculations of the composition factors of $({\rm
HS}{:}2)_{w_{420}}$ excludes the first possibility, namely $L_3(4){:}2^2$. Hence $({\rm HS})_{w_{420}} \cong S_8
\times 2.$

Similarly we can deduce that $({\rm HS}{:}2)_{w_{480}} \in \{2
\times A_6 \cdot 2^2\cdot 2,\; H, \; K, \; M, N \}$, where,
possibly,  $H$ is a subgroup of index $154$ in $M_{22}{:}2$, $K$ of
index $4$ in $2^5 \cdot S_6$, $M$ of index $14$ in $S_8 \times 2$
and $N$ of index $14$ in $L_3(4){:}2^2.$ We deal with the
elimination of $H$, $K,$ $M$ and $N$ in the following:

\begin{itemize}
\item[(i)] From the list of maximal subgroups of $M_{22}{:}2$, there are two
possible candidates for $H$, either a subgroup of index $7$ in
$L_3(4){:}2_2$ or of index $2$ in $2^4{:}S_6.$ The list of maximal
subgroups of $L_3(4)$ shows that it contains no subgroup of index
$7$. The group $2^4{:}S_6$ is a maximal subgroup of $M_{22}{:}2$ and
computations with Magma show that its non-trivial normal subgroups
are of type $2^4$, and hence it cannot have a subgroup of index $2$.
\item[(ii)]We constructed the maximal subgroup $2^5 \cdot S_6$ inside ${\rm HS}{:}2$
and found out that it does not contain a subgroup of index $4$.
\item[(iii)]Lists of maximal subgroups of $S_8 \times 2$ and $L_3(4){:}2^2$
(see \cite {atlas}) eliminate the possibilities of $M$ and $N$.
\end{itemize}
 Therefore $({\rm HS}{:}2)_{w_{480}}=2 \times A_6 \cdot 2^2\cdot 2.$

Further, we can deduce that $({\rm HS}{:}2)_{w_{484}} \in
\{M_{22}{:} 2,\, A\}$, where,  possibly,  $A$ is a subgroup of index
$11$ in $L_3(4){:}2^2$ or $S_8 \times 2,$ or $A$ is a subgroup of
index $154$ in $2 \times A_6 \cdot 2^2\cdot 2.$ A careful
verification of each case rules out all other possibilities except
$M_{22}{:}2$. Hence we deduce that $({\rm HS}{:}2)_{w_{484}} \cong
M_{22}{:}2.$

Similarly by using the composition factors we deduce that $({\rm
HS}{:}2)_{w_{500}} \cong 5^{1+2} {:}(Q_8{:}4)$.

For $l =672$, we argue similarly as in the case $l =420$,  since
$A_{672} = A_{420}$. Thus, we deduce that $({\rm HS}{:}2)_{w_{672}}
\in \{L_3(4){:}2^2,\; S_8 \times 2, \:B\}$, where $B$ possibly is a
subgroup of $M_{22}{:}2$ of index $11$. Since $M_{22}{:}2$ contains
no subgroup of index $11$ we deduce that $B$ is either a subgroup of
$L_3(4){:}2^2$ or a subgroup of $\; S_8 \times 2.$ An examination of
the composition factors of $({\rm HS}{:}2)_{w_{672}}$ excludes the
second possibility, namely $S_8 \times 2$. Hence $({\rm
HS}{:}2)_{w_{672}}=L_3(4){:}2^2.$


It follows by the above case by case analysis that $({\rm
HS}{:}2)_{w_{420}},\,({\rm HS}{:}2)_{w_{480}}, \,({\rm
HS}{:}2)_{w_{484}},\, \\ ({\rm HS}{:}2)_{w_{500}}$ and $({\rm
HS}{:}2)_{w_{672}}$ are all maximal subgroups of ${\rm HS}{:}2.$

Now, by the transitivity of ${\rm HS}{:}2$ on the code coordinates,
the codewords of $W_l$ form a $1$-design $\CD_{w_l}$ with $A_l$
blocks. This implies that ${\rm HS}{:}2$ is transitive on the blocks
of $D_{w_l}$ for each $w_l$ and since $({\rm HS}{:}2)_{w_l}$, for $l
\in \{420, 480, 484, 500,\, 672\}$  is a maximal subgroup of ${\rm
HS}{:}2,$ we deduce that ${\rm HS}{:}2$ acts primitively on
$\CD_{w_l}$ for $l \in \{420, 480, 484, 500,\, 672\}.$ Note that
$\CD_{w_{420}},\,\CD_{w_{480}},\,\CD_{w_{484}},\,\CD_{w_{500}},\text{~and~}
\CD_{w_{672}}$ are $1$-designs with parameters  $1$-$(1100, 420,
420),$\,\, $1$-$(1100, 480, 6720)$,\, $1$-$(1100, 484, 44)$, \,
$1$-$(1100, 500, 10080),\,$ and  $1$-$(1100, 672, 672)$ respectively
with 1100, 15400, 100, 22176, and 1100 blocks.

Finally for $l=512,$ $W_{512}$ splits into two orbits of lengths
3850 and 4125, namely $W_{(512)_1}$ and $W_{(512)_2}$ respectively.
Let $u = w_{(512)_1} \in W_{(512)_1}$ and $v = w_{(512)_2} \in
W_{(512)_2.}$ Then $({\rm HS}{:}2)_u$ is a subgroup of order 23040,
and from the right hand side of TABLE~1 we deduce that $({\rm HS{:}2})_u \cong 2^5 \cdot
S_6.$ Similarly $|({\rm HS}{:}2)_v|$ = 21504  and $({\rm HS}{:}2)_v$
is a maximal subgroup of ${\rm HS}{:}2$ isomorphic to $4^3{:}(L_3(2)
\times 2).$ Notice that $D_u$ is a $1$-$(1100, 512, 1792)$ design
having 3850 blocks, while $D_v$ is a $1$-$(1100, 512, 1920)$ design
with 4125 blocks. ${\rm HS}{:}2$ acts primitively on $\CD_u$ and
$\CD_v$.~\end{proof}


\begin{lemma}\label{notmaxims}
Let  $l \in \overline{L}$ and $w_l \in W_l.$ Then $({\rm
HS}{:}2)_{w_l}$ is a non-maximal subgroup of ${\rm HS}{:}2.$
\end{lemma}

\begin{proof}We give a description of the cases $l =532,$ $l =544$ and $l
=548$ since the sets of codewords of these weights split into a
number of orbits. The remaining cases, i.e., $l =564, 576,$ and 612
are much simpler to be dealt with using similar arguments. Let
$l=532$. Then $W_{532}$ splits into two orbits of lengths 61600 and
132000, namely $W_{(532)_1},$ and $\,W_{(532)_2}$ respectively. Let
$a = w_{(532)_1} \in W_{(532)_1}$ and $b = w_{(532)_2} \in
W_{(532)_2.}$ Then $({\rm HS}{:}2)_a$ is a subgroup of order 1440,
and thus not maximal in ${\rm HS}{:}2$. Using  the composition
factors of $({\rm HS}{:}2)_a$ and the information in
\cite{hs2sublattice} and \cite{connor} we deduce that $({\rm HS})_a \cong
A_6{:}2{:}2.$ Similarly $|({\rm HS}{:}2)_b|$ = 672 and $({\rm
HS}{:}2)_b$ is a non-maximal subgroup of ${\rm HS}{:}2$ isomorphic
to $L_2(7){:}2{:}2.$

For $l=544$ we have that $W_{544}$ splits into three orbits of
lengths 77000, 346500 and 369600, namely $W_{(544)_1},\,W_{(544)_2}$
and $W_{(544)_3}$ respectively. Set $x = w_{(544)_1} \in
W_{(544)_1},x' = w_{(544)_2} \in W_{(544)_2}$ and $x'' = w_{(544)_3}
\in W_{(544)_3}.$
We used Magma and  \cite{hs2sublattice}, and also the information on
maximal subgroups of ${\rm HS}{:}2$, to determine the structure of
$({\rm HS}{:}2)_x$ and deduce that $({\rm HS}{:}2)_x=(({\rm
HS})_x){:}2 \cong (2^4{:}(S_3 \times S_3)){:}2.$ Similarly, since
$|({\rm HS}{:}2)_{x'}| = 256,$ and so not a maximal subgroup of
${\rm HS}{:}2$. We determined that $({\rm HS}{:}2)_{x'} \cong X{:}2$
where $X=((((4 \times 2){:}2){:}2){:} 2){:}2,$ and $X{:}2\leq P{:}2$
with $P{:}2 \in Syl_2({\rm HS}{:}2)$. We can easily show that
$$P{:}2\cong ((4.2^4){:}D_8){:}2\cong 2^{1+6}_+{:}D_8.$$
Clearly, $({\rm HS}{:}2)_{x''} \cong S_5{:}2.$

If $l=548$, then $W_{548}$ splits into two orbits of lengths 38500
and 462000, namely $W_{(548)_1},$ and $\,W_{(548)_2}$. Let $s =
w_{(548)_1} \in W_{(548)_1}$ and $t = w_{(548)_2} \in W_{(548)_2.}$
Then $({\rm HS}{:}2)_s$ is a subgroup of order 2304, and thus not
maximal in ${\rm HS}{:}2$. Using  the composition factors of $({\rm
HS}{:}2)_a$ and the information in \cite{hs2sublattice} we deduce
that $({\rm HS})_s \cong 2^4{:} S_3 {:} S_3 {:} 2 \times 2.$
Similarly $|({\rm HS}{:}2)_t|$ = 192
 and $({\rm HS}{:}2)_t$ is a non-maximal subgroup
of ${\rm HS}{:}2$ isomorphic to $2^{1+4}{:}S_3.$

Using similar arguments for $l = 564, 576$ and $612$ we deduce that
$({\rm HS}{:}2)_{w_{564}} \cong (2^3 \times S_3){:}S_3,$ and $({\rm
HS}{:}2)_{w_{576}} \cong (2^3{\cdot} S_4){:}2,$ and $({\rm
HS}{:}2)_{w_{612}} \cong ((2^4{:}A_5){:}2){:}2.$ ~\end{proof}

\medskip

TABLES 4 and 5 below, list the structures of $({\rm HS}{:}2)_{w_l}$
and ${\CD}_{w_l}$ for all $l$, respectively.
\subsection{Observations}\label{sec-obs}
\begin{itemize}

\item[(i)] In TABLE~4 the first column represents the codewords of
weight $l$ and the second column represents the stabilizer in ${\rm
HS}{:}2$ of a codeword $w_l$ of $W_l$. In the final column we test
the maximality of $({\rm HS}{:}2)_{w_l}$ in ${\rm HS}{:}2.$

%\pagebreak

{\footnotesize
\begin{center}
TABLE~4\\ Stabilizer in ${\rm HS}{:}2$ of a word $w_l$\\[1ex]
\begin{tabular}{lrc||lrc}
\hline \hline $l$  & $\qquad\qquad\quad\;\;\, ({\rm HS}{:}2)_{w_l}$ & Maximality & $l$  & $\qquad\qquad\quad\;\;\, ({\rm HS}{:}2)_{w_l}$ & Maximality\\
\hline
420  &  $S_8 \times 2$ & Yes & $(544)_2$  & $2^{1+6}_+{:}D_8$   & No\\[1ex]
480  &  $2\times A_6.2^2$ & Yes & $(544)_3$  & $S_5 {:} 2$  & No\\[1ex]
484  &  $M_{22}{:}2$ & Yes & $(548)_1$  &  $2^4{:} S_3 {:} S_3 {:} 2 \times 2$ & No\\[1ex]
500  &  $5^{1+2} {:}(Q_8{:}4)$ & Yes & $(548)_2$  &  $2^{1+4}{:}S_3$ & No\\[1ex]
$(512)_1$  &  $2^5 \cdot S_6$ & Yes & $564$  & $(2^3 \times S_3){:}S_3 $  & No\\[1ex]
$(512)_2$  &  $4^3(L_3(2) \times 2)$ & Yes & $576$  & $(2^3{\cdot} S_4){:}2$  & No\\[1ex]
$(532)_1$  &  $A_6{:}2{:}2$ & No & $612$  & $((2^4{:}A_5){:}2){:}2 $  & No\\[1ex]
$(532)_2$  &  $L_2(7){:}2{:}2$ & No & $672$  &  $L_3(4):2^2$ & Yes\\[1ex]
$(544)_1$  & $(2^4{:}(S_3 \times S_3)){:}2$  & No\\[1ex]
\hline \hline
\end{tabular}
\end{center}}

\item[(ii)] In TABLE~5 the first column represents the codewords of
weight $l$ and the second column gives the parameters of the designs
$\CD_{w_l}$ which were constructed in Section~\ref{stabhs2}. In the
third column we list the number of blocks of $\CD_{w_l}$. We test
the primitivity for the action of ${\rm HS}{:}2$ on $\CD_{w_l}$ in
the final column.

\pagebreak
{\scriptsize
\begin{center}
TABLE~5\\ $1$-designs $\CD_{w_l}$ from  ${\rm HS}{:}2$\\[1ex]
\begin{tabular}{llrc}
\hline \hline $l$  & $ \quad\;\;\, \CD_{w_l}$ & No. of blocks & Primitivity \\
\hline
420  &  $1$-$(1100, 420, 420)$ & 1100 & Yes \\[1ex]
480  &  $1$-$(1100,480,6720)$& 15400 & Yes \\[1ex]
484  &  $1$-$(1100, 484, 44)$ & 100 & Yes \\[1ex]
500  & $1$-$(1100, 500, 10080)$ &  22176 & Yes \\[1ex]
$(512)_1$ &$1$-$(1100, 512, 1792)$  & 3850 & Yes \\[1ex]
$(512)_2$  & $1$-$(1100, 512, 1920)$ &  4125 & Yes \\[1ex]
$(532)_1$ &$1$-$(1100, 532, 29792)$  &  61600 & No \\[1ex]
$(532)_2$  &$1$-$(1100, 532, 63840)$ &  132000 & No \\[1ex]
$(544)_1$  &$1$-$(1100, 544, 38080)$ & 77000  & No\\[1ex]
$(544)_2$  &$1$-$(1100, 544, 171360)$ &  346500  & No\\[1ex]
$(544)_3$  &$1$-$(1100, 544, 182784)$ &  369600   & No\\[1ex]
 $(548)_1$  & $1$-$(1100, 548, 19180)$ & 38500 & No\\[1ex]
$(548)_2$  & $1$-$(1100, 548, 230160)$ & 462000 & No\\[1ex]
 $564$  & $1$-$(1100, 564, 157920)$ & 308000 & No\\[1ex]
$576$  & $1$-$(1100, 576, 120960)$ & 231000  & No\\[1ex]
$612$  & $1$-$(1100, 612, 12852)$ &23100 & No\\[1ex]
$672$  &  $1$-$(1100, 672, 672)$& 1100 & Yes\\[1ex]
\hline \hline
\end{tabular}
\end{center}}
\end{itemize}

\section{Binary codes from the complementary
design}\label{compdescode}

It is often of interest to know whether a given code contains the
all-one vector. We showed in Proposition~\ref{thecodecode} that
$\jvec \in C^{\perp}.$ Since $\jvec \notin C$ we know that $C \neq
\tilde{C}$, where $\tilde{C}$ is the code of the complementary
$1$-$(1100, 680, 680)$ design $\tilde{\CD}$. In
Proposition~\ref{codecompl} below we collect the properties of
$\tilde{C}$. Observe by the weight distribution that $C$ and
$\tilde{C}$ are complementary codes.


\begin{proposition}\label{codecompl}
Let $\tilde{C}$ be the binary code defined by the incidence matrix
of the design $\tilde{\CD}$. Then $\tilde{C}$ is a self-orthogonal
doubly-even $[1100,21,480]_2$ code. Its dual code
${\tilde{C}}^\perp$ is a $[1100,1079,4]_2$ with words of weight $4$.
Furthermore, {\rm Aut}$(\tilde{C}) \cong {\rm HS}{:}2$.
\end{proposition}
\begin{proof}The proof follows similar arguments to those used in the proof
of Proposition~\ref{thecodecode}. So we omit the details.~\end{proof}
\begin{remark}\label{wdcodecomp}

%\pagebreak

The weight distribution of $\tilde{C}$ is listed in TABLE~6.

\begin{center}
{\footnotesize
TABLE~6: Weight distribution of $\tilde{C}$\\[1ex]
\begin{tabular}{lr||lr}
\hline \hline $l$ & $\qquad\qquad\quad\;\;\, A_l$ & $l$ & $\qquad\qquad\quad\;\;\, A_l$\\
\hline
0    & 1    & 568 &  193600 \\[1ex]
480  & 15400  & 576 &  231000 \\[1ex]
488  & 23100 & 600  & 22176  \\[1ex]
512 & 7975 &  616 &  100  \\[1ex]
536 & 308000 & 672 &  1100 \\[1ex]
544 & 793100 & 680 &  1100  \\[1ex]
552 & 500500 &  & \\[1ex]
\hline \hline
\end{tabular}}
\end{center}

A closer examination of TABLE~3 and 6 shows that the codewords of
$C$ and $\tilde{C}$ appear in complementary pairs. Hence, an
analysis of the structures of the stabilizers, their maximality and
the primitivity of the corresponding designs can be dealt with in a
manner similar to that in the previous results.
\end{remark}

\section{Concluding remarks}
The codes $C$ and $\tilde{C}$ meet in their doubly-even self-orthogonal code $C_0$. It turns out that $C_0$ is isomorphic to the code constructed in \cite{moorod8}.
$C_0$ consists just of the code vectors of $C$ whose weights are divisible by $32$. Let $J = \la \jvec \ra$ denote the repetition code generated by the all $1$-vector $\jvec$. Then $C_1 = C_0 + J$ is a self-orthogonal doubly-even  $[1100,21,428]_2$ code which is isomorphic to the code of the complementary $1$-$(1100, 428, 428)$  design discussed in \cite{moorod8}. We note that $C$, $\tilde{C}$ and $C_1$ are ${\rm HS}$-invariant subcodes of $C_2 = C + J$ containing $C_0$ with codimension 1.

\section*{Acknowledgments}
The authors would like to thank Jonathan Hall for suggesting this problem  in his review MR2884789 (2012m:05082). The authors also thank the anonymous referee for helpful and
constructive remarks and suggestions.



\end{document}
